Author's response to reviews

Title: A new approach to treatment of resistant gram-positive infections: Potential impact of targeted IV to oral antibiotic switch on length of stay.

Authors:

Mohammed Desai (mdesai@hhnt.nhs.uk)  
Kathleen B Bamford (K.Bamford@imperial.ac.uk)  
Sarah Trust (Sarah.Trust@bsuh.nhs.uk)  
Bryony D Franklin (bdean@hhnt.nhs.uk)  
Mike Richards (Mrichards@hhnt.nhs.uk)  
Ann Jacklin (ajacklin@hhnt.nhs.uk)  
Alison Holmes (Alison.holmes@imperial.ac.uk)

Version: 3  Date: 7 June 2006

Author's response to reviews: see over
We have now made all the requested formatting changes to the manuscript:

**Formatting changes requested**

Affiliations - Each affiliation should be written in full and be provided in the following format: department, institute, city, country. This may mean repeating institution addresses if authors are from different departments within an institution. Please do not use full capitals for country names, and please do not include full-stops in abbreviated country names.

>> We have reformatted each author’s affiliations to your specifications. Furthermore we have made minor changes to the departmental links of a few authors. The order of the authors listed has also been amended at the request of the lead author (Kathleen B Bamford). This has all been revised as follows:

Mohammed Desai\(^2\), Bryony Dean Franklin\(^2,5\), Alison H Holmes\(^1,4\), Sarah Trust\(^2\), Mike Richards\(^4\), Ann Jacklin\(^2\), Kathleen B Bamford\(^1,3\).

1 Department of Infectious Diseases and Microbiology, Imperial College, London, UK
2 Pharmacy Department, Hammersmith Hospitals NHS Trust, London, UK
3 Department of Microbiology, Hammersmith Hospitals NHS Trust, London, UK
4 Department of Infectious Diseases, Hammersmith Hospitals NHS Trust, London, UK
5 The School of Pharmacy, University of London, London, UK

References. The reference list should contain all authors' names, regardless of how many there are. The term ‘et al.’ should not be used.

>> We have now provided a comprehensive list of authors in all of our references. The term “Et al.” has been replaced in all appropriate references with the remainder of the authors for each reference now listed.
Tables - Table 1 has been divided into a and b. It is important that tables are numbered in ascending numerical order - 1, 2 and 3. Can we please ask that the tables are renumbered and any references to the tables in the text are updated to reflect these changes.

>> Table 1a has been renumbered 1, table 1b has been renumbered 2, and all other tables have been renumbered in ascending numerical order. All references in the text to tables have been changed to reflect the new number allocation.

We trust that these changes have now reformatted the manuscript to your requirements.

Kathleen B Bamford